\documentclass[12pt,twoside]{article}

\usepackage{xspace}

\newcommand{\coursenumber}{CS 6104}
\newcommand{\coursetitle}{Symbolic Computation}
\renewcommand{\docdate}{May 29, 1996}
\newcommand{\doctitle}{Scribes (Revised)}

\newcommand{\firstpageheading}[1]{
\vbox to 0pt{\vskip -#1
\hbox to \hsize{\sl \vbox to 8.5pt{}\doctitle\hfill\docdate\hfill \rm 1}\vss}
\nointerlineskip
\vspace*{-15pt}
}

\pagestyle{myheadings}
\markboth{\hfill\docdate\hfill\doctitle}{\doctitle\hfill\docdate\hfill}

\addtolength{\textwidth}{1.50in}
\addtolength{\textheight}{1.00in}
\addtolength{\evensidemargin}{-1.25in}
\addtolength{\oddsidemargin}{-0.25in}
\addtolength{\topmargin}{-.50in}

\begin{document}

\firstpageheading{50pt}

\thispagestyle{empty}

\begin{center}
\Large\bf\doctitle\\
\smallskip
\coursenumber---\coursetitle\\
\smallskip
\Large\bf First Summer Session, 1996\\
\end{center}

Here is the list of scribe assignments.
Please let me know if you cannot be scribe on any of your assigned dates.
Scribe notes are due by email two classes following the class
that is the source for the notes.
I may return notes to you for revisions.
The notes will be distributed on the class web site and as paper handouts.

\begin{center}
\begin{tabular}{|c|c|l||c|c|l|}
\hline
\sc Class & \sc Date & \sc Scribes &
\sc Class & \sc Date & \sc Scribes \\\hline\hline
1 & 5/20 & Craig &
15 & 6/10 & Scott, Ted\\\hline
2 & 5/21 & Ben, Heather &
16 & 6/11 & Craig, Julie\\\hline
3 & 5/22 & Luoqi, Matt, Scott &
17 & 6/12 & Ben, Heather\\\hline
4 & 5/23 & Mike, Rob &
18 & 6/13 & Luoqi, Matt, Scott\\\hline
5 & 5/24 & Gabriel, Ted &
19 & 6/14 & Mike, Rob\\\hline
6 & 5/28 & Craig, Julie &
20 & 6/17 & Gabriel, Ted\\\hline
7 & 5/29 & Ben, Heather &
21 & 6/18 & Craig, Julie\\\hline
8 & 5/30 & Luoqi, Matt, Scott &
22 & 6/19 & Ben, Heather\\\hline
9 & 5/31 & Mike, Rob &
23 & 6/20 & Luoqi, Matt, Scott\\\hline
10 & 6/3 & Gabriel, Ted &
24 & 6/21 & Mike, Rob\\\hline
11 & 6/4 & Craig, Julie &
25 & 6/24 & Gabriel, Ted\\\hline
12 & 6/5 & Ben, Heather &
26 & 6/25 & Craig, Julie\\\hline
13 & 6/6 & Luoqi, Matt, Scott &
27 & 6/26 & Ben, Heather\\\hline
14 & 6/7 & Mike, Rob &
28 & 6/27 & Luoqi, Matt, Scott\\\hline
\end{tabular}
\end{center}

\end{document}
